Author's response to reviews

Title: Mast Cell Regulation of Na-Glutamine Co-Transporters B0AT1 in Villus and SN2 in Crypt Cells during Chronic Intestinal Inflammation.

Authors:

Soudamani Singh (singhs@marshall.edu)
Subha Arthur (arthursu@marshall.edu)
Jamil Talukder (Jamil-Talukder@loc.edu)
Balasubramanian Palaniappan (Palaniappan@marshall.edu)
Steven Coon (coons@bu.edu)
Uma Sundaram (sundaramu@marshall.edu)

Version: 4  Date: 7 February 2015

Author's response to reviews:

Editor's comment: -This paper is acceptable, pending minor edits to fix grammatical errors in the revision. In the tracked version, in the new text, there are multiple errors of missing articles and lack of subject-verb agreement. I suggest that the corresponding author go over the manuscript additions and send in a corrected version.

Response: As recommended by the Editor, we have now made all the requested changes. Thank you.

Uma Sundaram, MD: